Author's response to reviews

Title: An evaluation of the integration of non-traditional learning tools into a community based breast and cervical cancer education program: the witness project of Buffalo

Authors:

   Dr Thelma C. Hurd (thelma.hurd@roswellpark.org)
   Prof Paola Muti (muti@buffalo.edu)
   Deborah O. Erwin (erwindeboraho@uams.edu)
   Sharita Womack (sdwomack@buffalo.edu)

Version: 7 Date: 27 May 2003

The references have been renumbered as requested in the discussion and reference sections of the manuscript.